ELENA and AD commissioning
status

2329.0ms

IA F

AL
§MCycle BHZ Cycle ‘_}

Dw 3q_ms Rr:09

521

|

. . ; EH?

pirmd W W N

ar-2021 10:38: 54 oA

l HMPROD?2 HMMDOPT l}]l]]EEI HMMDOPT HMPROD2 HMMDOPT -}]]]]EEI

.‘
I

I
!

10:38:54
10:38:46
10:38:40
10:38:34
10:38:28
10:38:20

Tpu 3. Oe‘f'

HMMDOPT
HMMDRF
HMMDOPT
HMPROD2
HMMDOPT
HMMDRF

|Kicker | |BTV15 |BTV117 |BTV118 |Scrapers| |[B | E /S | Tl1lalL |- | @

Tpu X, 1e‘5

Tpu 1.6e6

“ALPHA QATRAP2 ASACUSA2
QALPHA

QALPHA QATRAP2 QASACUSA2

QALPHA
@ALPHA QATRAP2 QASACUSA2

OALPHA

x2.3 -

@BASE
@BASE

@BASE

Tpu: 1 6e5

Tpu e® 2.31+0.66-Nn:28%
Tpu e® 1.05+0.00-n: 0%
Tpu e® 3.30=0.84-n:25%
Tpu e® 0.75=0.00-n: 0%
Tpu e°1.58=0.32-n:19%

Tpu 2.3e®

Lpu Lne0O | Lpu Lne50
2.151e® 8.374e°
0.194e° 4,593e°

0805.821e® 2.235e°
0.255e® 7.039e®
©0.330e® 14.982e®

)

ELENA



©) Highligh of the week

- ELENA transfer line:
— Vacuum OK in LNEO2 and LNEOb
— electrostatic elements connected according to new optics in
LNEO2 — Powering tests validation on-going

— Missing electronics of the profile monitors for the special H one
in AEGIS zone

~ Cycle accelerating and decelerating the beam:
— Continue working on improving setting of working point
— Orbit correction along the cycle, test of e-cooler compensation knobs
— Work on-going to improve injection efficiency
— Continue testing new RF functionalities

— Synchronization problem on scrapers instrument to be followed-up with
STI and BI

~HW commissioning in AD
— Vacuum leak in stochastic cooling kicker repaired with resin
— Access system interlock cable repaired
 DSO tests planned for 16/04
— Repair/re-installation of magnet interlock cables on-going

Note: W16 access in ELENA on Wednesday 21 April (TS in injectors) instead of Tuesday
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